OPEN ACCESS **FODDOTICS** ISSN 2218-6581 www.mdpi.com/journal/robotics

Review

Psychophysiological Methods to Evaluate User's Response in Human Robot Interaction: A Review and Feasibility Study

Lorenza Tiberio^{1,2,*}, Amedeo Cesta¹ and Marta Olivetti Belardinelli^{2,3}

- ¹ CNR—Italian National Research Council, ISTC, Via San Martino Della Battaglia 44, Rome 00185, Italy; E-Mail: amedeo.cesta@istc.cnr.it
- ² ECONA (Interuniversity Centre for the Research on Cognitive Processing in Natural and Artificial Systems), Rome 00185, Italy; E-Mail: marta.olivetti@uniroma1.it
- ³ Department of Psychology, Universit àdegli Studi di Roma "La Sapienza", Rome 00185, Italy
- * Author to whom correspondence should be addressed; E-Mail: lorenza.tiberio@istc.cnr.it; Tel.: +39-06-4459-5221; Fax: +39-06-4459-95243.

Received: 24 March 2013; in revised form: 13 May 2013 / Accepted: 14 May 2013 / Published: 10 June 2013

Abstract: Implementing psychophysiological measures is a worthwhile approach for understanding human reaction to robot presence in terms of individual emotional state. This paper reviews the suitability of using psychophysiological assessment in human-robot interaction (HRI) research. A review of most common psychophysiological parameters used in a controlled laboratory setting is provided and advantages and challenges of their utilization in HRI experiments are described. Exemplar studies focused on the implementation of psychophysiological measures for the evaluation of the emotional responses of the participants to the robots' presence are described. Based on the reviewed literature, the paper also describes the results of our own research experience to make the most of the emerged recommendations. We planned and performed a study aimed at implementing psychophysiological measurements for assessing the human response of two groups of older adults (Healthy *vs.* Mild Cognitive Impairment subjects) towards a telepresence robot. Finally, the paper provides a summary of lessons learned across the field in using psychophysiological measures in HRI studies.

Keywords: psychophysiological measurements; human-robot interaction; human response; socially assistive robots

1. Introduction

Humans generally interact with robots in the same way they might interact with other people, establishing social relationships and emotional ties with them [1–4]. One of the main characteristics relative to assistive social robotics is that a robot proactively engages users in a social manner creating an interaction with the person for the purpose of giving assistance and support in certain activities of daily living and care [4–6]. Gross *et al.* [7] report that the application of assistive social robots for domestic use has gone one step further towards the introduction of robotics technologies in real homes.

As assistive social robots are being designed to become a part of our life, an extremely important issue regards the development of methods to assess user's experience with a robot, while understanding how humans feel during their interaction with it [8]. A robot operating at home, in nursing care or in a rehabilitation unit is required to have determined social capabilities, like natural language communication skills or emotion expression and/or perceptual skills. Such features, in fact, maintain and facilitate the interaction with human users [9,10]. How people respond and react to the robot's social skills, and in general, to other features such as its functionality, role and appearance, is an important aspect of social interaction, which affects the acceptance and use of the robot. In addition, it is difficult to imagine social interactions occurring without emotional responses and therefore the analysis of the interactions between a human and a robot should be based also on a clear understanding of the human's reaction to the robot in terms of emotions. An emotional response against a stimulus, such as the daily interaction at home with a robot, implies an involvement of the central nervous system (CNS) through multiple cortical and subcortical regions and an activation (in terms of arousal) of the autonomic nervous system (ANS), which can be registered by physiological measurements [11–14]. In brief, the emotional states can be categorized along two main dimensions, one that explains the valence of the emotion (in a continuum of pleasantness-unpleasantness) and another that refers to the corresponding level of arousal/physiological activation. According to the "circumplex model of affect" [15], every emotion can be explained as a linear combination of valence (positive or negative) and intensity of activation (arousal). An additional dimension (dominance/control) has been frequently suggested for emotions with the same level of arousal and valence (e.g., fear and anger), but that are distinguishable in everyday life [16].

In this paper, the main focus is on the psychophysiological measurements of human response to a robot. Psychophysiological measures are considered one of the main methods of assessment used for human studies in human robot interaction (HRI) together with self-report, behavioral measures and task performance analysis, as Bethel *et al.* report in [8]. In this survey, the authors provide a detailed overview of the process of designing and planning user evaluation based on a psychophysiological approach applied to detecting and/or identifying users' emotions, modifying in real-time robot control and behavior, as well as evaluating the user's reactions to technology. The aim pursued in this review is to expand the knowledge on the assessment users' reactions to technology by means of psychophysiological measures as reported in [8]. In the scope, we have taken into account recent researches aimed at determining how humans react in terms of physiological response during an interaction with a robot in order to establish the real effectiveness of human-robot interaction. The comprehension of these aspects will encourage users to collaborate and share the environment with a robot designed, for instance, to provide them with social, health-care and rehabilitative services.

The paper is organized as follows: Section 2 briefly presents a description of psychophysiological terminology along with a survey of applications in the human-computer interaction research. Advantages and disadvantages of using psychophysiological measures are also reported. Section 3 focuses on the most common psychophysiological measures used in human-robot interaction studies in a controlled laboratory setting. Section 4 provides an overview of the most recent studies in the field of HRI that apply psychophysiological methods to evaluate the human's reaction to a robot in terms of affective-emotional response. Section 5 presents our feasibility study focused on the assessment of the human response to a telepresence robot in a group of healthy older adults and in a group of older patients suffering from mild cognitive impairment (MCI). Section 6 summarizes the results of both the reviewed literature and of our study, leading to the final discussion.

2. Psychophysiology for Human Studies

Citing from [17]: "Psychophysiology concerns the study of cognitive, emotional and behavioral phenomena as related to and revealed through physiological principles and events". Through the monitoring and interpreting of physiological signals during the interaction between a human user and an assistive robot, it is possible to obtain information about the user's inner experience in terms of affective and cognitive states [8,18,19].

Methods based on psychophysiological assessment have been previously used in human computer interaction (HCI) to evaluate users' cognitive workload [20], emotional responses [21], and cognitive-emotional states [22] during interaction with a technology. A significant increase of electromyography activity and hearth rate, and a significant decrease in blood volume pulse were found in research focused on individual's response to audio and video degradations in videoconferencing software [23]. Scheirer et al. [24] used electromyography activity and blood volume pulse to distinguish subjects' frustration during the utilization of a pre-programmed mouse delay. Particularly, the psychophysiological game evaluation measures the effects that games have on the players [25], how people respond to game features [26], users' experience and emotional engagement [27]. Mandryk et al. [28] analyze user experience with entertainment technology. Participants in their experiment report higher electromyography and electrodermal activity when playing against a friend than against a computer. In another study, Mandryk and Atkins [22] evaluate different methods for modeling emotion using physiological data through two fuzzy logic approaches. The first fuzzy logic model transformed electrodermal (GSR), electromyography (smiling and frowning) and cardiovascular activity (HR) into arousal and valence. The second model transformed arousal and valence into five emotions. Their results prove the reliability of physiological metrics to model emotional user's experience with interactive play technologies.

Psychophysiological measures have been more recently used in HRI human studies [8] in combination with subjective research methods, as behavioral and self-report measurements, in order to achieve a more complete and reliable knowledge of the person's experience. Generally, when a person communicates his/her own personal opinion about a research issue, as in our case about the interaction with a robot, this may be affected by mood and/or social desirability. In addition, observer bias may have an effect on participants' responses and consequently reactions or behaviors may not reflect the exact feeling of the persons involved in an evaluation study of HRI [8,29].

The psychophysiological measures offer numerous advantages and disadvantages that are well identified in the literature [8,18,30,31]. One of the main benefits of using these techniques is the objectivity of the assessment since participants generally cannot manipulate their own physiological reactions. Moreover, psychophysiological monitoring is a non-invasive method that is able to detect continually and in real time different cognitive and affective states when they occurs in response to a stimulus. However, the use of psychophysiological techniques poses also different challenges related to the modality of data acquisition and interpretation. Moreover, the quality of the recording may be influenced by both confounding environmental factors, such as noise or lightning, and by individual psychological internal state during the evaluation that might undermine the reliability and the correct interpretation of the gathered data. Physiological monitoring requires the use and management of specialized equipment. The accurate placement of electrodes on the body of the participant is essential for a correct acquisition of the parameters. Generally the monitoring of these measures occurs in a laboratory setting where the naturalness of the task or the interaction is lacking and the participants' response may be biased.

Wherever designing a study aimed at the evaluation of any human-system interaction by means of psychophysiological measurements, specific guidelines should be taken into account in order to gather useful data for a precise analysis of the user's experience [32]. Every experimental design should consider a set of phases. First of all, a researcher should select the appropriate psychophysiological parameters to be employed in order to identify definite cognitive and affective user's states during the interaction with a robot. It is necessary to determine the sample size and characteristics. After determining the experimental setting (laboratory or real-world) and equipment, a pre-testing assessment should be performed with the purpose of verifying potential confounding factors. Participants should be informed about the aim and the procedure of the study through a consent form to be signed. Prior to the experimental evaluation, the researcher should gather some health and psychological information of the participants. Prior and following the assessment, the experimental design should also consider the administering of other tools, such as questionnaires/interviews or video-recording, to acquire additional information that might help the interpretation of the psychophysiological signals and of the users' experience. The gradual adaptation to the stimulus, or in this case to the interaction with a robot, could modify participants' answers. Finally, a debriefing phase should be conducted to reflect on the experience and make it meaningful by identifying any issue that may have had an effect on the participant's response and behavior.

These guidelines are well discussed in Bethel *et al.* [32], and although the psychophysiological approach can be useful in human-robot interaction analysis in order to assess participants' reactions and responses in terms of affective and cognitive states, in the experimental design, a researcher should always bear in mind advantages and disadvantages of the use of psychophysiological measures. The evaluation of the participants' experience should focus on different methods assessment in order to gain a more complete understanding of the aspects that affect the quality of the interaction between a service robot and a human.

The next section illustrates the psychophysiological measures mostly applied in HRI.

3. Common Psychophysiological Measures in HRI

The nervous system can be divided into two parts: the central nervous system and the peripheral nervous system. The peripheral nervous system is subdivided into the sensory-somatic nervous system and the autonomic nervous system. As mentioned in the introduction, the central and autonomic nervous system responses can indicate possible modifications in our emotions, motivation, attention, and preferences. Without going into details, it can be stated that a system of multiple cortical (e.g., frontal, temporal, and parietal) and subcortical (e.g., basal ganglia, thalamus, amygdala, and hippocampus) regions contributes to the experience of emotion. Each cerebral area has a different role and a specific contribution to the elaboration of the stimulus [12]. According to right-hemisphere and valence hypothesis, this cerebral hemisphere is dominant for emotional processing, and in particular, for decoding negative emotions differently from left-hemisphere specialized for positive emotions [33,34]. The ANS is part of the peripheral nervous system (PNS) containing all the nerves that lie outside the central nervous system (CNS). The key role of the PNS is to connect the CNS to the organs, limbs and skin and to adapt organisms' internal state to the internal and environmental demands maintaining homeostasis. The ANS is further divided into two branches: the sympathetic system regulates the "flight-or-fight" responses, while the parasympathetic system helps to maintain normal body functions and conserves physical resources. An activation of ANS is also recognized as consequence of emotional responses [35]; negative emotions seem to activate a stronger autonomic reaction than the positive ones. The functioning and activation of the ANS can be assessed through psychophysiological parameters. Recently, a psychophysiological approach has been adopted to assess the internal state of subjects interacting with the technology. In this section, we briefly introduce some widely used psychophysiological measures applied in human-robot interaction studies.

3.1. Cardiovascular Activity

The main measures of cardiovascular activity are heart rate (HR), blood pressure (BP), blood volume pulse (BVP) and heart rate variability (HRV). HR is the number of heartbeats per unit of time-typically expressed as beats per min (bpm). In terms of autonomic responding, increases in sympathetic activity are associated with heart rate rate rise, and relative increases in parasympathetic activity are associated with heart rate reduction [36,37]. The force exerted by blood on the walls of blood vessels is the blood pressure. Systolic blood pressure (SBP) is a measure of blood pressure when the ventricles are contracting and the heart beats. Diastolic blood pressure (DBP) is a measure of blood pressure, while on the contrary, the parasympathetic system decreases cardiac activity reducing blood pressure [38]. HRV is the sequence of time intervals between heart beats. This interbeat interval time series is used to calculate the variability in the timing of the heart beat in response to changing physiological conditions and factors such as respiration rate or emotional states. Consequently, HRV is a clear effect of regulation by the sympathetic and parasympathetic nervous system. Sympathetic activity is associated with the low frequency components (LF) while parasympathetic activity is associated with the higher frequency components (HF) of modulation frequencies of the HR [39].

Other cardiovascular measures include respiratory sinus arrhythmia (RSA) defined as the natural variation in heart rate that is primarily driven by breathing patterns and its regulating influence of sympathetic activity on the heart. Although the role of RSA in emotional response and regulation is not yet clear, according to the literature, its response measures might be a reliable index of emotional valence in situations demanding emotion regulation [40,41]. Inhaling inhibits the activity of the sympathetic system, increases heart rate, and decreases HRV. Exhaling activates the sympathetic branch, decreases heart rate, and increases HRV.

Several studies have investigated the relationship of cardiovascular indicators with arousal and cognitive load [42,43]. Cardiovascular response is a good marker of emotional states of the subjects [35,44–46]. An increase in HR, SBP, DBP and in respiratory activity, but not in HRV, has been reported in anger, anxiety and happiness responses [40,47,48].

3.2. Electrodermal Activity

Electrodermal activity (or galvanic skin response or skin conductance) is based on the variation of the skin electrical resistance in response to various emotional stimuli. This variation is mainly due to the level of sweat of the skin as a result of the action of the eccrine glands [49–51]. Sweat glands are primarily driven by sympathetic innervations. There are two main components of electrodermal activity:

- (1) the tonic component (Skin Conductance Level) is the absolute value of the skin conductance at a given moment in absence of a specific stimulus onset. So-called nonspecific skin conductance responses also occur in absence of specific (individual) emotional stimuli, with their frequency proportional to arousal.
- (2) the phasic component (Skin Conductance Response) is the response to an emotional stimulus.

Electrodermal activity is measured placing two electrodes on the skin surface of two adjacent fingers and passing a tiny electric charge. When the subject responds to a stimulus, the skin immediately becomes a slightly better conductor of electricity. External emotional stimuli produce a drop in the skin electrical resistance and in particular at the level of the palm and soles of the feet.

Electrodermal activity is related to the level of arousal elicited by a wide range of psychological and emotional states with either positive or negative valence. Different studies investigating anxiety, anger, fear and also joy experience report increased electrodermal activity [52,53]. Electrodermal activity is also an indicator of the cognitive load, stress and arousal [42,51,54,55].

3.3. Electromyography Activity

Electromyographic signals (EMG) are produced by the activation of the muscle motor unit and provide an index of the muscle activity. A muscle is composed of a multitude of motor units that are activated by electrical signals elicited by the nervous system. When a motor unit is activated, an electrochemical impulse is fired and propagates through the nerve fibers. Along the nerve fibers, the impulse stimulates the skeletal muscle creating muscle contraction, and consequently, the movement of human limbs [56]. Surface EMG is a method of recording muscle activity by electrodes mounted directly on the skin. Surface EMG is evaluated by means of the analysis of the signal amplitude in microvolt. The muscles more frequently monitored are those reflecting the human emotional state and

that can serve to assess the emotional regulation (frontalis, corrugator supercilii, zygmaticus major, levator labii superioris, orbicularis oculi) [11,57]. The facial electromyography gives information related to the affective state of subjects. A rise of zygomatic major muscle activity results in a happy facial stimuli, whereas people tend to contract corrugators supercilii in presence of angry facial stimuli [35,58,59]. The zygomatic muscle lifts up the lips to produce a smile, whereas the corrugator muscle knits the eyebrows forming a frown facial expression. Dimberg *et al.* [60] show that electromyographic responses to positive and negative stimuli can be generated without any conscious awareness of the emotional eliciting stimuli.

3.4. Brain Activity

Psychophysiological reactions based on the central nervous system activity can provide information about the brain responses to emotional stimuli. Electroencephalography (EEG) is a technique that records the brain electrical activity through the placement of electrodes over the surface of the scalp. The electrodes record the rapidly fluctuating of brain wave patterns, which are a sign of continuous electrical changes in the brain. EEGs show a different involvement of the two hemispheres of the brain (cerebral asymmetry) in responding to positive and negative emotional stimuli. A higher activation of the left hemisphere indicates a positive reaction, whereas the right hemisphere is sensitive to negative affect [35,61]. Specifically, prefrontal cortical regions of the brain are involved in emotional responding. Several studies show that frontal EEG asymmetry is considered a mediator of emotional processing [62]. Some of the earliest studies about the frontal EEG asymmetry and emotions were done by Davidson *et al.* [61], and concerned differences in hemispheric activity during the viewing of emotional films. Results showed that more active right hemisphere activity was related to negative content in films while more left hemisphere activity was related to happy films. Recently, evidence suggests a relationship between brain activity and anxiety [63]. Distinct patterns of brain activity are associated with different types of anxiety. For example, anxious apprehension involves an increased left-hemisphere activity and anxious arousal is related to a greater right-hemisphere activity [64]. Electroencephalography activity can also be used as a measure of mental workload. Chaouachi et al. [65] present a workload index based on features extracted from EEG signals. Wilson [66] classifies operators' workload level by taking EEG as an input and reports results of 90% accuracy.

Functional magnetic resonance imaging (fMRI) is a further method used to evaluate human response during the interaction with a robot. This technique displays the hemodynamic response (changes in blood flow and oxygenation) related to neural activity in the brain or spinal cord. When nerve cells are active, they consume oxygen carried by hemoglobin. The effect of this consumption is an increase in blood flow in regions where there is a greater brain activity. When measuring changes in blood flow associated with increasing neuronal activity, it is possible to map the activation of brain areas during an activity or in response to a stimulus. Specific brain areas (amygdala, medial prefontral cortex, anterior cingulated cortex, insula) are supposed to be involved for emotional processes [67]. Saygin *et al.* underline the value of an interdisciplinary collaboration between robotics and cognitive neuroscience to improve the understanding of how robots' actions, behavior and embodiment are perceived and processed in the human brain [68]. Some relevant research has focused on Theory of Mind (ToM) [69]. When we interact with each other, it is possible to infer the interlocutor's internal

state, observing behavior, ways of speaking and facial expressions. The other's internal state refers to their intentions, goals and beliefs, on the basis of which, it is possible to interpret the behavior of others. Brain imaging methods, such as fMRI have been widely applied also to identify brain areas for ToM in order to explore how human users respond, perceive and represent robots' actions, behavior and embodiment during an interaction [70].

4. Psychophysiology Applied in HRI Studies: An Overview

As mentioned in the introduction, Bethel *et al.* identify three main categories of research studies implementing psychophysiological measurements in HRI [8]. The first is focused on detecting emotions expressed by human users, based on the development of control architecture, behavior system or online emotion detection algorithms able to recognize, classify and interpret participants' affective state [19,70–80]. The second category is related to the analysis of subjects' psychophysiological responses for development and implementation of real-time robot control architectures, and adaptation of robot behaviors in order to, for instance, maximize users' motivation, engagement and performance during rehabilitation therapy [64,81–86]. The third category is the objective of this review, and it comprises research studies aimed to evaluate human experience and emotional reaction to the technology. The intent in the next subsection is to review the most relevant and recent works specifically oriented to using psychophysiology measures in understanding humans' emotional response and reactions towards a robot. A separate subsection describes a set of recent studies implementing functional magnetic resonance imaging (fMRI) as a method to evaluate how the human brain reacts to the robot's features and behavior.

4.1. Psychophysiological Studies Evaluating Humans' Response to Robots

In the field of research that implement psychophysiological measures to assess how humans react to a robot, Kulić and Croft evaluated the feasibility of physiological measures as indicators of human subjective response towards motions of a robotic manipulator arm [72]. Electrodermal (GSR), electromyography (corrugators muscle) and cardiovascular activity (HR) of 36 participants were measured during an experiment in which the human response to two different robot motions was assessed. The robot could be attracted to the target location and be rejected avoiding obstacles (potential field planned robot motions) and it could have the capability to avoid obstacles (safe planned robot motions). Two different tasks were performed by the manipulator robotic arm: (1) pick and place (2) reach and retract. Three different speeds for the manipulator arm were provided for each task.

Psychophysiological reactions were analyzed using a fuzzy inference engine as an estimation method for the relationship between physiological responses and emotional categories. Fuzzification is the process of converting the input variables' values into a set of membership values in the interval $\{0,1\}$ in the corresponding fuzzy sets [87]. Physiological signals are not measured directly via the fuzzy inference engine but they are preprocessed. The relevant features extracted are used for hypothesis and for establishing a set of rules base to estimate the human emotional state. In this research, the fuzzification was derived using data from psychophysiological research [17].

Psychophysiological data were compared with subjective responses based on a Likert scale developed to assess anxiety, calm and surprise levels of the participants. Results showed a correlation

between anxiety and speed, and surprise and speed of the robotic manipulator arm. In addition, subjects tended to show a stronger response in electrodermal, electromyography and cardiovascular activity (in terms of higher levels of anxiety and surprise and lower levels of calm) for potential field planned motions than safe planned motions. An estimate of arousal and valence of human emotional state was obtained from the fuzzy inference engine, but it has not proven a capability to successfully estimate valence.

In a second study, researchers developed a new human affective state estimator based on the hidden markov model (HMM). They performed the same experimental method and procedure in [72] and compared the HMM with the fuzzy inference engine. Their results showed that HMM is a better affective state estimator, able to identify the relationship between physiological signals and valence and arousal [19].

Dehais *et al.* [88] evaluate human response to different types of robot motions. They compare different ways to move a robotic arm close to the "human space". Twelve participants were recruited in order to evaluate the effects to three different robot motion types (*Motion 1.* with planner, with grasp detection, medium velocity; *Motion 2.* no planner, no grasp detection, high velocity; *Motion 3.* with planner, no grasp detection, low velocity) on physiological parameters, such as electrodermal activity, deltoid and ocular activities. The task consisted of taking a bottle from the robot's grip and participants were then asked to report their opinion on a nine-point visual analog scale, reporting the legibility of the different types of robot motions, safety and physical comfort perceived during the experiment. Their results, similar to those of Kulić and Croft in [72], show a higher electrodermal activity for high velocity robot motion compared to the other two motions characterized by medium or low velocity. The participants interpreted the electrodermal activity with a higher level of arousal and anxiety in Motion 2. The lowest deltoid and ocular activity (in terms of fixation time) was found for Motion 1. with planner, grasp detection and medium velocity.

In the results of the above studies, a common aspect lies in the belief that by combining different measurements (physiological parameters and self-report measures), it is possible to obtain information on the nature of the interaction between a human and a robot, and on the specific human reaction. The results show that the robot's behavior has an important effect on the user's reactions and it can influence acceptance toward a robot and the user's perception of interaction safety.

Human physiological responses are also involved in the assessment of specific user's responses to different tasks performed through a robot. Goljar *et al.* [89] recruited three different groups of subjects: a subacute stroke (N = 23 subjects), a chronic stroke (N = 10 subjects) and a control group (N = 23). Electrodermal, cardiovascular activity and respiratory rate were monitored during a reaching and grasping exercise task performed through a manipulator robotic arm and a haptic robotic interface. The aim of the study was focused on understanding if stroke patients' physiological responses were different from the responses of healthy subjects in relation to a physically undemanding task, and which physiological responses were most reliable in evaluating psychological states in motor rehabilitation training. Two different types of each were presented to the subjects (virtual task *vs.* harder virtual task required to catch a ball before it reached the lower end of a virtual table. The manipulator guided the subjects' arms in order to move left or right and reach the ball. In the harder virtual task, subjects performed the same exercises but the robot had inverted left-right

controls. Physiological data was also compared with 9-point arousal and valence scales from the Self-Assessment Manikin, in order to have an index for the participants of their level of emotional valence and arousal during the experiment. Results showed that the electrodermal signal appears to be the most valuable for obtaining useful information about participants' arousal state. Specifically, electrodermal activity of the control group was higher than the subacute and chronic stroke groups during both types of task. A significant correlation between electrodermal activity and Self-Assessment Manikin was also found in all three groups. Beside some limitations of the study discussed in the paper, results confirmed the utility of selecting appropriate psychophysiological measures to get supplementary information about users' psychological state and the efficacy of robots in clinical practice.

Also in this study, feedback obtained from the user's physiological parameters is useful for evaluating the emotional status of a human who performs a rehabilitation task through a robot. Individual reactions in rehabilitation practices may have an effect on the patient's performance that could be affected by the robot's behavior. The efficacy of human-robot interaction may depend also on individual motivation and engagement so it is important to ensure that the robot's behavior is able to adapt to a user's state, needs and capabilities.

Recently, Swangnetr and Kaber [90] recruited 24 participants at a senior center in order to evaluate the relationship between physiological response and emotional states, and to develop a machine-learning algorithm for their accurate classification. Heart Rate (HR) and electrodermal activity were collected during a simulated medicine delivery experiment in which participants interacted with a PeopleBot robot. The experimental design counted three independent variables: robot facial feature (abstract *vs.* android); speech capability (synthesized *vs.* digitized voice; and different modes of user interaction (visual messages *vs.* physical confirmation of receipt of medication with a touch screen). All participants were exposed to each variable. At the end of each experimental session (14 test trials), subjects were asked to complete the Self-Assessment Manikin in order to indicate their emotional response to the different robot patterns. In general, respect to the variables on the robot feature type, the results showed a significant difference in terms of an increased HR and electrodermal activity to the condition in which the robot required physical confirmation of receipt of medication with a touch screen. Referring to the development of a machine-learning algorithm, for HR and electrodermal activity features extraction, a wavelet analysis was used with the purpose of classifying human emotional state in real time.

Results showed that HR was a stronger index of valence compared to electrodermal activity, while arousal states had a strong relationship with electrodermal activity. The wavelet analysis was confirmed as a valuable method for identifying associations between physiological response and arousal and valence. A correct classification of human emotional state in response to a robot in a healthcare setting could contribute to ensure efficacy and quality of the human-robot interaction.

Similar results in Zhang and colleagues [91] showed that interactivity features of a service robot stimulated higher cardiovascular and electrodermal response in 24 older participants during a set of experimental trials in which a robot executed a medicine delivery task. Participants' cardiovascular (HR) and electrodermal activity was collected during the experimental trials during the interaction with the robot. A fuzzy logic model was adopted in order to gather physiological data in terms of arousal and valence. The objective of the study was to understand how different service robot interfaces (face, voice messaging and user interactivity) affect elderly people's perceptions and emotional response.

The robot had an abstract or human-like face and a synthesized or digitized human voice. User interactivity consisted in two levels of interaction based on the reading of a visual message and the request to confirm medicine delivery. Participants indicated a more positive emotional experience interacting with a service robot with anthropomorphic features based on interactivity aspects.

Broadbent and colleagues [92] explored if mental schemas about robots' humanness were associated with their psychophysiological reactions to a robot. They recruited 57 adult participants and asked them to draw their idea of a healthcare robot before seeing a real robot (PeopleBot), explaining how to attach the blood pressure cuff for measuring it. Blood pressure was monitored at baseline and then during the interaction with a robot. Participants completed also the positive and negative affect scale (PANAS) [93]. An increase in blood pressure, as an index of the ANS, was registered in participants who had drawn the robot more human-like compared to those that imagined a more machine-like robot when they interacted with the real robot. In addition, participants who drew a robot compared to those who drew a machine-like robot.

A possible interpretation of these results may refer to the high responsiveness to stress and arousal by the ANS. Assessment and interpretation of human physiological responses towards features and behavior of a robot provide valuable elements for understanding the degree of applicability of a robot based on the context of use, on the type of task, on interaction modality and on users' needs and preferences.

4.2. fMRI Studies Evaluating Humans' Responses to Robots

A further technique to assess the human reaction during the interaction with a robot is found in the study of brain response. The assessment of brain activity is an interdisciplinary area involving different disciplines, as cognitive neuroscience, psychology, robotics engineering and social science, and gaining a great interest in the field of HRI studies. Saygin et al. [68] recruited 20 participants in a fMRI experiment in which they viewed video clips of three agents carrying out actions, such as drinking water from a cup or waving their hand. The three agents differed in form: human; robot with human form; robot with nonhuman form. The aim of the study was to explore whether the human brain would show a different specialization area for human form and motion. Researchers expected similar brain responses for human and robot with human form, and differentials for robot with nonhuman form, while for motion they supposed similar responses for robot with human form and robot with nonhuman form, and differentials for human. Results of this study demonstrated that robot with human form was processed in a different brain area compared to both human and robot with nonhuman form. Specifically they found an activation in the parietal cortex for the robot with nonhuman form. The researchers supposed a prediction error based on the incongruity between the human-like form and the mechanical movements that the brain is not able to integrate. Chaminade et al. [94] used fMRI as a method to confirm the hypothesis of a preference in autistic patients for a small humanoid robot during a computerized computer game. They recruited two experimental groups of subjects: 18 healthy students and 12 participants with Autism diagnoses. All subjects participated in the rock-paper-scissors game in a computerized version. Participants were told that they would play against three different opponents: the experimenter, an artificial agent and a generator embedded in a computer. The subjects knew that the first two opponents (active opponent) playing strategies used to win. The fMRI data were collected during the three game sessions. Results confirmed that autistic patients considered the artificial agent as a human opponent since the same level of brain activity in the temporal area of autistic patients was registered by means of fMRI when they interacted with the robot and with the human. In another study [95], fMRI was used to test the hypothesis of a reduced brain activity in areas involved in motor and emotional processing during the interaction with a robotic agent compared to human agent. The experimental paradigm consisted of 13 healthy young participants and 16 experimental conditions in which human and robot agents impersonated four facial expressions (anger, joy, disgust and speech). Participants observed the human and robot agents performing the gestures by way of short videos while their brain activity was scanned. An increased activity of occipital and posterior temporal regions was found in response to the gesture of robotic agent compared to the human agent, while a response in insula and orbitofrontal cortex respectively for the perception of disgust and anger emerged only for human gesture.

In order to determine the brain human response to the robot's embodiment during an interaction, Miura *et al.* [96] analyzed fMRI data of 39 students during an interaction with bipedal or wheel drive humanoid robots. A human like embodiment of the robot would involve a greater emotional response in terms of brain activity compared to a robot with a more machine-like embodiment. The experimental procedure consisted in six conditions in which a human, a bipedal robot (more human-like) and a wheel drive humanoid robot (more machine-like) performed emotionally positive and neutral daily life actions (e.g., walking happily). The fMRI scanning was along the six conditions. Results revealed that the left orbi-frontal cortex was more active during the interaction with the bipedal humanoid robot performing emotionally positive actions compared to the wheel drive humanoid robot. It is worth noting how the same brain response was found during the interaction with the human.

The studies described above confirm that the psychophysiological approach is well applicable to assess emotional response and quality of experience during human robot interaction. The benefits from the use of physiological parameters in HRI research are well documented by [8].

5. Using Psycho-Physiological Measures to Evaluate Older Adults' Response to a Telepresence Robot: A Feasibility Study

To ensure an effective human-robot interaction, it is necessary to understand also how end users react to the presence of a robot and to its features (e.g., abilities, appearance, behavior). Given the mentioned benefits of psychophysiological approach and drawing inspiration from the reviewed literature, we designed a study aimed to implement psychophysiological measurements for assessing the human response towards a telepresence robot (results in this section partially appeared in [97]). Telepresence robots have been more and more often proposed to be used to enhance older adults' social and daily functioning and to supplement human care-giving. Robots, as in Giraff [98], Texai [99], and Kompai [100], have been put forward as a proposed form of assistive and communicative device, able to deliver health and social services. To date, literature has not focused on empirical studies that employ a psychophysiological approach specifically for human-telepresence robot interaction. Our intent was to explore the impact of a telepresence robot in terms of emotional human response in older adults. We have relied on the use of psychophysiological measures presuming also that this approach

may offer valuable information also about potential problems in the human-robot interaction, that need to be taken into account in order to make the proposed technology more usable, satisfactory and adaptable to the user's needs. The objective of the study was to examine the effectiveness of a remotely controlled telepresence robot, called Giraff [101], as a tool for health and social services. Subjective and psychophysiological measures were collected in order to evaluate older users' responses towards robots' presence and to obtain initial feedback about the suitability of a telepresence robot to support older adults.

5.1. Apparatus

Giraff is a social robotic telepresence system manufactured by the Giraff Technologies AB (Figure 1). It is a human-height physical robot integrated with a videoconferencing system mounted on a mobile robotic base. A user (called *pilot*) can remotely connect through an Internet connection to the robot via a computer interface and can move around in the environment where the robot is located and interact with other humans (called *local users*). Giraff is part of the ExCITE European Project [102] dedicated to study the use of robotic telepresence as a means to promote social interaction and inclusion of older adults [98].





The experimental setting was composed of two rooms. In the first room participants interacted directly with the human experimenter, whereas the second room was used by the experimenter to interact through the telepresence robot. The experimenter maneuvered the robot through the pilot from the second room and met the subject remotely.

5.2. Physiological Parameters

The physiological monitoring during the study was performed through a Holter monitor. Cardiovascular activity in terms of heart rate (HR) and heart rate variability (HRV) was measured. As described in the previous sections, HR and HRV result from the activation of the ANS, which consists of two components: the sympathetic and the parasympathetic systems. Based on this assumption, an increase in sympathetic activity was expected, associated with heart rate rise, as well as relative increases in parasympathetic activity, associated with slower heart rate [4]. HRV is a consequence of regulation by the sympathetic and parasympathetic nervous system. The most frequently used HRV components are spectral parameters such as low-frequency (LF) band and high-frequency (HF) band percentual powers. Respectively, the LF band is considered as markers of the sympathetic nervous system, while the HF band reflects parasympathetic (vagal) activity A same pattern of activation of ANS was expected both in the interaction between human experimenter and participants and in the interaction with the robot. The protocol should have included an analysis of the collaboration between the parties involved in the research, have limited the study only to the analysis of HR and HRV. We do not exclude the possibility of expanding the results by analyzing other physiological measures in later stages of the study.

5.3. Subjective Measures

The administered self-report measures were:

- The *State-Trait Anxiety Inventory* [103] (STAI). It is a self-report questionnaire consisting of two scales (Y1 for state anxiety and Y2 for trait anxiety) containing 20 items each. All items are rated on a 4-point scale (*i.e.*, from 1 = almost never, to 4 = almost always).
- (2) The *Positive and Negative Affect Schedule* [93] (PANAS). It consists of two scales containing 10 items each, one measuring positive affect and the other measuring negative affect. Each item is rated on a 5-point scale (*i.e.*, from 1 = very slightly or not at all, to 5 = extremely).
- (3) A concluding interview on the robot experience was administered in order to get information about: **social presence** ("*I had the feeling that the investigator was in the same place where I was*"), **utility and advantages of the robot** ("*I think that Giraff could be useful in the home*"), **engagement** ("*I felt physically involved in the experience with Giraff"*), **physical aspect of the robot** ("*Giraff is pleasing to see"*), **privacy issues** ("*It may be annoying to receive virtual visits through the Giraff at home"*).

5.4. Participants

Seventeen participants were recruited at the rehabilitation department of the Fondazione "Don Gnocchi" of Rome. Given the intent to validate the use of the telepresence robot as an aid for health and social services, it was decided to focus on a specific type of older users, recruiting a group of elderly with a diagnosis of mild cognitive impairment (MCI) and a group of healthy older adults. The elderly group with a diagnosis of MCI was or had been engaged in cognitive rehabilitation interventions at the Fondazione. The characteristics of the sample are as follows:

(1) Normal control group (NC), which consisted of 9 participants (age 70 \pm 4.3 years, age range 65–75, 6 males, 3 females) with no cognitive impairment. Subjects were tested with the Mini-Mental State Examination (MMSE) reporting a score of 30. Mean score on Trait Anxiety scale (Y2) was 32.3 (with *Spielberger's recommended cut-off of 39/40* [82]).

(2) MCI group, which consisted of 8 participants (age 73.5 \pm 5.2 years, age range 65–79, 5 males, 3 females). MCIs were assessed by a battery of standardized neuropsychological tests and their score on MMSE was between 25 and 30 (mean 27 \pm 1.18). Mean score on Trait Anxiety scale (Y2) was 34.5 (with *Spielberger's recommended cut-off of 39/40* [82]).

5.5. Procedure

The study has received the consent by the Ethical Committee. On the basis of agreements with representative health professionals of the Fondazione, experimental procedure took place over the course of four-day laboratory sessions performed at the Fondazione "Don Gnocchi". Upon arrival to the first-day laboratory session, participants were asked to read the consent form and to sign it. They were also asked to fill the STAI-Y2 and a demographics form. Participants were asked to avoid nicotine, caffeine, alcohol, other substances, and strenuous exercise at least three hours before their scheduled laboratory session. The experimental procedure is show in Figure 2 and it was articulated into four main sessions in which the participants interacted:

Figure 2. Experimental procedure.



- (1) directly with the human experimenter during the first two sessions;
- (2) through the telepresence robot during the subsequent two sessions.

Two sessions for each type of interaction were scheduled in order to allow participants to know the experimenter and the robot (the sequence of the sessions was established according to the agreement with the limitations imposed by the Foundation for the patient's safeguard). Each interaction session had three main phases. During the *Baseline* (T0), HR and HRV assessment at rest was performed. Baseline was followed by an *Interaction Phase* (T1) in which the experimenter administered cognitive stimulation tasks to the participant. Finally a Recovery Phase (T2) was monitored for regaining HR and HRV steady state. Physiological assessment was continuous during each phase of each session. During the laboratory session the participant was seated in front of the experimenter and the Holter electrode was applied to the subject's chest.

As indicated by the data literature of human studies in HRI measures, a timing between 2 to 5 min is reported to compute cardiovascular parameters [64,84,91]. In line with this indication, and according to the healthcare professionals of the Fondazione, we chose a duration of 3 min for the *baseline* phase,

during which participants were asked to rest. Following the Baseline Phase (T0), participants were engaged in 10 min of *Interaction Phase* with cognitive stimulation (T1). Following the interaction phase, participants were asked to remain seated for 3 min (*recovery* phase, T2). STAI-Y1 and PANAS scale were administered at the end of the recovery phase of the first and third session, in order to gain feedback in terms of anxiety and positive/negative affects for both types of interaction (human and robot). At the end of the fourth day laboratory session, participants met the experimenter again for the final interview focused on their experience with the telepresence robot.

Cognitive Stimulation Tasks

Cognitive stimulation tasks were usual activities for the participants of our experiment. We have decided to employ some tasks similar to the cognitive exercises of the cognitive rehabilitation interventions of the Fondazione to which patients had been or were engaged previously. This choice was purposeful also for examining the human/robot interaction in a rehabilitation context.

Particularly, in accordance to the cognitive therapist of the Fondazione, we have chosen to administer the following cognitive exercises:

- Word list memory: a list of 10 words is read to the subject at a constant rate of 1 word every 2 s. The word list is presented 3 times to the subject. At the end of each of the 3 presentations, the subject is asked to recall the list of words.
- (2). Verbal fluency test: subject is asked to name in one minute a list of words within a specific semantic (e.g., vegetables or professions) or phonetic category (words that begin with letter F, A, and S).
- (3). Story recall: the participants are asked to memorize the story once and then to complete an immediate recall test, at the end of the Interaction Phase.
- (4). Digit span Forward / Backward: in the forward digit-span task a series of numbers (e.g., 7–3–9) are presented to the subjects and they have to immediately repeat them back in the given order. If they do this successfully, they are given a longer list. In the backward digit-span task the participant needs to reverse the order of the numbers.
- (5). Numerical and abstract reasoning: in the first case participants are asked to resolve simple operations in addition, subtraction, multiplication or division; for the abstract reasoning subjects have to interpret some popular proverbs.

5.6. Results

For statistical significance, we used a multivariate analysis of variance (MANOVA) with one *between* subjects factor (*Group*: NC or MCI) and three *within* subjects factor (*Presence_Type*: human or robot; *Session*: first or second observation; *Phase*: baseline, interaction and recovery). A comparison of the HR variables in all phases of the experiment showed no significant difference in the baseline values.

5.6.1. Heart Rate

As we supposed before, an increase in sympathetic activity associated with heart rate rise was expected in response to the cognitive stimulation activity. Results showed a main effect for the variable *Phase*, $F(_{2.28}) = 35.7$, p < 0.05, $\eta^2 = 0.71$, indicating that heart rate was significantly higher during the *interaction* (T1) with respect to the *baseline* (T0) for both groups. A significant interaction *Phase* x *Group* (NC or MCI), $F(_{2.28}) = 5.26$, p < 0.05, $\eta^2 = 0.27$, indicated a higher increase in HR at T1 in NCs compared to MCIs (means are reported in Figure 3). A tendency to significant interaction was found for *Presence_Type* and *Group* $F(_{1.15}) = 3.79$, p < 0.07, $\eta^2 = 0.20$, indicating a higher HR in NC group compared to MCI during participants-experimenter direct interaction. There were no significant *Presence_Type* and *Session* effects in HR response during the various phases suggesting that there is a comparable cardiovascular reaction in terms of HR both in the participants-experimenter direct interaction and in the robot mediated interactions. A possible explanation of this result might lead to consider the change in Heart Rate in NCs as an index of a higher arousal in response to the cognitive stimulation activity independently on the type of presence (human or robot).

Figure 3. Mean for HR for phase (T1, T2, T3) for both groups during the interaction with human and telepresence robot.



5.6.2. Heart Rate Variability

The mean of low frequency (LF) and high frequency (HF) were considered for each phase of each experimental session. We expected a predominance of the LF band as a marker of sympathetic nervous system activation and a decrease in HF band as result of parasympathetic in response to cognitive stimulation activity during the interaction both in the participants-experimenter direct interaction and in the robot-mediated interactions. A significant interaction *Phase* x *Presence_Type* (human or robot), $F(_{2.28}) = 4,18$, p < 0.05, $\eta^2 = 0.21$ was found. This result indicated that both groups showed a higher increase in LF percentual power at T1 during the interaction with the robot. There was a significant interaction *Phase* × *Presence_Type* × *Group*, $F(_{2.28}) = 4,6$, p < 0.05, $\eta^2 = 0.23$. Both groups showed a decrease for HF percentage power at T1. A significant effect for *Presence_Type* was found in both

groups but the MCIs showed a higher decrease of HF percentage power at T1 during the robot-mediated interaction with respect to the NC group. A possible explanation might lead to consider these results as an indicator of a higher stress level in MCIs in response to a cognitive stimulation activity during the robot-mediated interaction.

5.6.3. Subjective Measures

State Trait Anxiety Inventory (STAI): no significant differences were observed between groups in terms of state anxiety (p = 0.33). In addition, no major effects were observed with respect to the *Presence_Type* (human or robot). No significant correlation was found between state anxiety and physiological parameters.

Positive and Negative Affect Scale (PANAS): mean scores were calculated for participants-experimenter direct interaction and robot-mediated interaction. Mean higher scores on Positive Affect subscale were found in both groups for both *Presence_Type*. NC group's mean score for participants-experimenter direct interaction = 37.3, while mean score for robot-mediated interaction = 37.5; similarly MCI group's mean score for participants-experimenter direct interaction = 31.4. Comparison between the Positive and Negative Affect Scales score and physiological parameters collected indicates meaningful correlation for Negative Affect. We found that Negative Affect indicated higher significant correlations with the HR Phase at T1 (r = 0.545, p < 0.05) e T2 (r = 0.503, p < 0.05) during the first session of participants-experimenter direct interaction and T3 (r=.573, p < 0.05) of the second session of robot-mediated interaction. These results indicated an experience of negative emotions, especially for robot-mediated interaction.

Final interview: in general our participants affirmed they felt physically involved in the experience with the robot (%NC = 100 and %MCI = 87). The experience with the robot was judged as rather pleasant (%NC = 89 and %MCI = 75) and involving (%NC = 78 and %MCI = 62) and totally spontaneous. NCs and MCIs subjects have a sufficient perception of the experimenter, as she was in the same place where they were (%NC = 56 and %MCI = 62). In addition, 78% of NC and 62% MCI participants asserted they would feel safer with the robot at home. However, only 50% of MCI group in case of need would be useful at home for 89% of NCs compared with only 25% of MCIs. Most of the participants stated that the presence of the robot would not disturb their life (%NC = 78 and %MCI = 75) but 75% of MCI subjects, compared with 44% of the NC group, considered the robot as annoying as a possible visit from a person. Lastly, the robot was judged pleasant to see (%NC = 100 and %MCI = 75) but all the participants of NC group, compared with 25% of MCI, judged the robot too big.

5.6.4. Summary and Discussion of Results of the Study

The feasibility study demonstrated that the robot presence did not cause adverse effects in terms of cardiovascular response and these results support the possibility for using a robotic mediated interaction by a telepresence system in clinical practice and also for a specific target patient as older adults with MCI diagnosis. During the participants-experimenter direct and robot-mediated interactions, HR data

showed a pattern similar for the NCs and the MCIs group as a response to cognitive stimulation tasks. HR was higher during the *interaction* (T1) with respect to the *baseline* (T0) for both groups. However, the significant main effect of the Group on heart rate response showed higher mean heart rate for NC group than MCI group. The NC group's heart rate increases from baseline during the interaction period and this could be considered as an indicator of higher arousal. The increase of arousal was slightly greater in the NC group during participants-experimenter direct interaction.

In terms of ANS response, the MCI group showed a higher mean sympathetic activity (predominance of LF band) and a reduced parasympathetic activity (decreasing in HF band) compared to the NC group. The higher sympathetic activity dominance found in the MCI group could be an index of the MCIs higher mental stress during interactions mediated by the telepresence robot. Despite the MCI group exhibited significant more mental stress, they reported subjective anxiety scores similar to the NC group for participants-experimenter direct and robot-mediated interactions. A general positive attitude of the telepresence robot was reported by participants, and this aspect might influence older adults' perceptions of the usefulness and ease of use of robots [104].

Some limitations of our study are consistent with other studies mentioned in this review [68,88,90,91,94] and are worth being highlighted. First, the small sample size of this study could preclude the generalizability of our results. The recruitment of MCI patients has encountered several difficulties. For instance, patients were asked to come to the Fondazione, and this request represented more effort for participants' relatives. Second, a weak association between physiological responses and self-reported measures in the case of the MCI participants would require further investigation. Although there is a clear difference between the groups in HR response during the interaction phase, we may affirm that the presence of the robot has not caused anxiety in the participants. However, based on the correlation between the Negative Affect scale of PANAS, the robot's presence may have had a slight negative connotation in terms of affects. Third, the MCI's medications might have altered some physiological responses (especially for HR) despite the two experimental groups showing the same cardiovascular clinical pattern. Fourth, the indications for experimental procedure suggested by the representative healthcare staff of the Fondazione did not allow us to control possible habituation effects. Finally, the laboratory setting/experiment may have increased the task engagement in the NC group because they could be more motivated to achieve a satisfactory performance.

Notwithstanding the limitations of the study, our results confirm the efficacy of the psychophysiological approach in studying human robot interaction as a valuable method also for comparing the reactions of different user groups. Telepresence robots, such as Giraff, have the potential to guarantee continuity of care, allowing users to be video-monitored by caregivers or healthcare providers remotely. Before the implementation of a robot in a user's life environment, it is recommended to evaluate users' characteristics and their emotional reaction to the robot. A positive reaction will affect the quality of experience and the effectiveness of the interaction. Psychophysiological assessment is a viable method to understand how a patient with specific diseases can react to the robot's presence and to its features, offering valuable feedback to understand the feasibility of an assistance program based on a robotic device.

6. Some Lessons Learned from the Literature

Starting with the assumption that assistive social robotics has been increasingly proposed as a tool for health care and social services, especially for older adults and their caregivers [105], a very important issue must be to expand our knowledge about how humans may react and interpret the robot's behavior, robot-performed human-like gestures, the robot's ability to perform specific tasks, and how the robot's physical embodiment may affect an interaction. The psychophysiological approach is confirmed as a valuable method to investigate the human's response towards an interacting robot. This section is intended to provide a summary of the guidelines underscored above, re-examined in the light of the reviewed literature and of our study on the older adults' responses to a telepresence robot.

Psychophysiological measures and their detection: Cardiovascular, electrodermal, electromyography and brain activity are widely employed in the studies of this review and their results are in line with previous studies on physiology of emotion [11,35], confirming that electrodermal (GSR) and cardiovascular (HR) measures are the most commonly assessed indices of ANS activation and of stimulus valence and arousal activation [14,90,106,107]. Consequently, it is recommended to refer mainly to cardiovascular and electrodermal variables in order to obtain an autonomic assessment. Psychophysiological parameters should be suitable for the experimental environment's features (laboratory, virtual or field setting), for the participant's profile (children-young and old), and as already mentioned, for the specific emotional reaction to measures. Similarly, the selection of the most appropriate equipment for detecting variables of research interest have to be considered one of the strategies for the overall control of the research plan. In a psychophysiological assessment, it is important that the experimenter learn to manage the equipment for the detection of the physiological parameters. In addition, it is suggested that the experimenter be supported by neuropsychopathology staff for the collection and interpretation of data. Physiological equipment or devices are not always easy to wear, and some of them may be invasive, affecting the participant's behavior and reaction. It is recommended, whenever it is possible, to choose non-invasive devices that do not exceedingly obstruct the person's physical body movements. Regarding this aspect, a phase of equipment functioning pre-assessment, events timing, and acquisition of the data to be recorded, should be taken into account.

Robots and participants as part of the experiment: The participants involved in the studies of this review interacted and reacted to different robot's features regarding the behavior, the level of anthropomorphism and the type of task performed. These variables can affect the robot's safety and effectiveness of the interaction perceived in relation to the user needs and context of use. In [22,68,94] participants reported positive emotions in response to human-like robots, confirming the results of Minato *et al.* [108] based on the Uncanny Valley hypothesis [109]. Human-like robot's features in terms of gestures and movements generate a positive response among the participants and consequently affect their attitude and acceptance towards the robot. The most used robots are generally robotic manipulator arms [19,72,89] social assistive robots, such as PeopleBot [91,92], mechanical humanoid robots [68,94,96], and robotic haptic interfaces [89], with which participants do not have sufficient familiarity. As reported in the literature, user's previous experience and familiarity contribute to modulate acceptance, especially in older adults [110,104]. It is likely that a lack of familiarity may have an adverse effect on the user's reaction. In the recruitment of the sample it is useful to consider all the external variables (such as technology experience) as well as the users' characteristics that may

generally affect their attitude and performance [104]. Among users' characteristics, healthy status and psychological information should be evaluated at the beginning of the experiment. Clearly, the cardiovascular pattern will affect the participant's reaction and it will be taken into account for possible confounds. One point is worth being underscored: the subjects recruited for sampling in many of the HRI studies described in this review are university students or young adults (e.g., [84,89,111]). Furthermore, stroke victim subjects tend to be younger adults with an average age of 50 years. The average age of an elderly subject enrolled in several studies was maximum 69 years. Just two of the mentioned studies refer to a sample composed exclusively of older adults [90,91] with an average of 80 years. The sample size varies from a minimum of 12 [88] to a maximum of 57 participants [92]. In the research of the category of study considered in this review, the issue of sample size identified by Bethel persists [8]. However, in [8] Kulic et al. [72], they had the largest sample (with 36 individuals), and in this review Broadbent et al. research had the largest sample recruited (with 57 participants). Determining the sample size before collecting data is important in order to achieve sufficient significance from the results [32]. This is not always easy, especially when the sample, such as ours, includes people with a specific disease whose recruitment and participation in the experiment can be demanding. As suggested by Bethel [32], a power analysis would always be advisable to determine the sample size required to detect an effect of a given size.

Combining objective and subjective measures: The experimental method of the studies is based on the main recommendations in HRI research when using a psychophysiological approach [32,112,113]: (1) combining multiple psychophysiology measures in order to find correlations in the results; (2) employing psychophysiology parameters as a valuable complement of self-report measures. Among the most commonly used self-report measures administered, the Self-Assessment Manikin [16] is a very common assessment technique to measure the pleasure, arousal, and dominance associated with participants' emotional reaction to robot stimuli as reported in [89,90]. Other research includes self-report assessment developed ad hoc [19,72,88] or a scale for evaluating affective response as positive and negative affect schedule (PANAS) [93]. A multimodal approach can be convenient to facilitate the understanding of the human response based on the evaluation of the correlation between behavioral feedback/physiological indicators in response to the stimulus (*i.e.*, the robot) and what the participant reports about the experience with the robot itself. In addition, objectivity of psychophysiological methods is a resource to avoid social desirability effects because we know that self-report assessment is subject to sources of error and bias; the participant usually wishes to provide a "socially acceptable" answer rather than to report his/her true feelings [114]. A study protocol should also consider the habituation effect that is very common in psychophysiological studies. Habituation has several implications for the reliability and stability assessment of results produced by studies based on the psychophysiological approach. Administering self-report measures or asking participants to evaluate their experience in the presentation of each task or stimulus of the experiment [8] is a consolidation strategy to avoid or reduce the negative effects of habituation, as reported in [72,88–91].

Briefing and Debriefing: In the planning phase of a study designed to evaluate the effectiveness of human-robot interaction through emotional response, it is advisable to define a written study protocol including, in addition to the different phases of the research, a phase of initial briefing to give all the necessary instructions and to obtain consent forms from the participants, as well as a final debriefing phase. A debriefing phase is considered, for example, in [90] by means of a final interview performed

after the experiment. Debriefing is a worthwhile means to gather information about more significant aspects of the user's experience and to ensure that the experience did not have a negative impact on the participants. In this way it is possible to reduce the effect of possible confounds and to get more information for a correct understanding of the physiological data collected during the experiment.

Experimental setting: The laboratory setting remains the primary experimental environment, because the methods for physiological parameters assessment require a controlled setting where it is easier to use technical equipment and to prevent unexpected events that could affect participants' reaction. In addition, the laboratory setting allows experimenters to operate faster on any malfunction of the robot and on the equipment used for the experiment. One of the main disadvantages of a laboratory setting consists in the minor ecological validity of the results. Field experiments take place in the participant's natural environment and consequently their results can be more easily generalized. Therefore, whenever it is possible, the experimental setups should be as close as possible to the participants' real life environment or even so familiar, using known contexts that are representative of real life situations (as a clinic for cognitive rehabilitation) in which the participant is already used to environmental and organizational features.

The lessons learned from this review confirm the value of developing guidelines for both planning and conducting the experiments that implement psychophysiological assessment in HRI. We believe that pulling out guidelines from different experiences in this research field is the best strategy for defining standardized methods to maximize reliability and generalizability of the results.

7. Conclusions

This paper presented a literature review of studies conducted in the HRI field research focused on the psychophysiological assessment as an evaluation method for measuring human emotional reactions during the interaction with a robot. The interest lies in expanding our knowledge of how humans react and what their affective and emotional responses are when they interact and collaborate with a robot. The skills and features of assistive social robots represent an important aspect of human robot interaction which, involving human emotional response, may contribute to affect both the quality and effectiveness of the interaction itself. From the psychophysiological feedback, it is possible to assess both the valence and levels of arousal associated to the interaction experience between a human and a social assistive robot. According to a "circumplex model of affect", emotional responses occur from two fundamental neurophysiological systems, one related to valence (a pleasure-displeasure continuum) and the other to arousal [15,115]. An emotional response, such as stress, is modeled as high arousal and low valence, while joy would be modeled as high arousal and high valence. This kind of information may be helpful to clarify if the user interprets the interaction with a robot as a negative or a positive experience. Positive experience will facilitate human-robot interaction and encourage collaboration between robot and user. In a context of real utilization, the assessment of individual emotional responses, as well as the analysis of user needs, is essential to the implementation of a robotic device.

We reported significant studies in this research field, and we also included studies implementing fMRI technique to assess the feedback from brain activity relative to how human users respond, perceive and represent robots' actions, behavior and embodiment during an interaction. From the

analysis of this literature we derived the elements useful to contribute to the reinforcement and development of guidelines to standardized methods based on the psychophysiological approach.

Overall, HRI studies based on the psychophysiological approach can be demanding, but following a careful planning of the research and taking into account some useful recommendations, it is possible to obtain encouraging results. The value of the psychophysiological feedback provides a key factor in increasing the knowledge about how the human being reacts during interactions with robots. Using these type of measures, it is possible to explore issues like understanding: (i) how to facilitate human-robot interaction; (ii) which users are more suitable to interact with a particular type of social assistive robot; (iii) what is the adequate level of anthropomorphism; and (iv) what is the best behavior that the robot should adopt relative to its features, the user's characteristics and the task to be performed for the human. Based on this information, we believe that it will be possible to render the human-robot interaction increasingly effective. A successful interaction will have a positive impact on individual motivation, attitude and performance, and will encourage the user to collaborate with the robot. We believe that the emotional response assessment based on psychophysiological measures could be a helpful approach to match individuals with the most appropriate robotic device depending on their needs and use preferences.

Acknowledgments

Authors would like to thank: the Fondazione Don C. Gnocchi of Rome for the field environment for this research and for the precious support of the neuropsychopathology staff, under the direction of Luca Padua, in the data collection and management. CNR authors would like to acknowledge Gabriella Cortellessa for joint work on the topic and the ExCITE Project (EU AAL JP) for partially supporting this work.

References

- 1. Breazeal, C. Towards sociable robots. *Robot. Auton. Syst.* 2003, 42, 167–175.
- 2. Forlizzi, J.; di Salvo, C. Service Robots in the Domestic Environment: A Study of the Roomba Vacuum in the Home. In Proceedings of the 1st Annual Conference on Human-Robot Interaction, Salt Lake City, UT, USA, March 2006; pp. 258–265.
- 3. Reeves, B.; Nass, C. *The Media Equation*; CSLI Publications: Cambridge, UK, 1996.
- 4. Sung, J.Y.; Guo, L.; Grinter, R.E.; Christensen, H.I. My roomba is rambo: Intimate home appliances. *Ubiquitous Comp.* **2007**, *4717*, 145–162.
- 5. Broekens, J.; Heerink, M.; Rosendal, H. Assistive social robots in elderly care: A review. *Gerontechnology* **2009**, *8*, 94–103.
- Feil-Seifer, D.; Mataric, M.J. Defining Socially Assistive Robotics. In Proceedings of the IEEE International Conference on Rehabilitation Robotics, Chicago, IL, USA, 28 June–1 July 2005; pp. 465–468.
- Gross, H.-M.; Schröter, C.H.; Müller, S.; Volkhardt, M.; Einhorn, E.; Bley, A.; Langner, T.; Merten, M.; Huijnen, C.; van den Heuvel, H.; *et al.* Further Progress towards a Home Robot Companion for People with Mild Cognitive Impairment. In Proceedings of the IEEE International Conference on Systems, Man, and Cybernetics (IEEE-SMC 2012), Seoul, Korea, 14–17 October 2012; pp. 637–644.

- 8. Bethel, C.L.; Salomon, K.; Murphy, R.R.; Burke, J.L. Survey of Psychophysiology Measurements Applied to Human-Robot Interaction. In Proceedings of the 16th IEEE International Symposium on Robot & Human Interactive Communication, Jeju Island, Korea, 26–29 August 2007.
- 9. Dautenhahn, K. Socially intelligent robots: Dimensions of human-robot interaction. *Philos. Trans. Roy. Soc. Biol. Sci.* **2007**, *362*, 679–704.
- 10. Fong, T.; Nourbakhsh, I.; Dautenhahn, K. A survey of socially interactive robots. *Rob. Auton. Syst.* **2003**, *42*, 143–166.
- 11. Cacioppo, J.T.; Berntson, G.G.; Klein, D.J.; Poehlmann, K.M. The psychophysiology of emotion across the lifespan. *Annu. Rev. Gerontol. Geriatr.* **1997**, *17*, 27–74.
- 12. Hagemann, D.; Waldstein, S.; Thayer, J.F. Central and autonomic nervous system integration in emotion. *Brain Cogn.* **2003**, *52*, 79–87.
- 13. Oatley, K.; Keltner, D.; Jenkins, J.M. *Understanding Emotions*, 2nd ed.; Wiley-Blackwell Publishing: Oxford, UK, 2006.
- 14. Swangnetr, M. Analysis of Patient-Robot Interaction Using Statistical and Signal Processing Methods. Ph.D. Thesis, North Carolina State University, Raleigh, NC, USA, 2010.
- 15. Russell, J.A. A circumplex model of affect. J. Pers. Soc. Psychol. 1980, 39, 1151–1178.
- 16. Morris, J.D. SAM: The self-assessment manikin. An efficient cross-cultural measurement of emotional response. *J. Advert. Res.* **1995**, *35*, 63–68.
- Cacioppo, J.T.; Tassinary, L.G. Psychophysiology and psychophysiological inference. In *Principles* of *Psychophysiology*; Cacioppo, J.T., Tassinary, L.G., Eds.; Cambridge University Press: Cambridge, UK, 1990; pp. 3–33.
- 18. Dirican, A.C.; Gökt ürk, M. Psychophysiological measures of human cognitive states applied in human computer interaction. *Procedia Comput. Sci.* **2011**, *3*, 1361–1367.
- Kulić, D.; Croft, E. Affective state estimation for human-robot interaction *IEEE Trans. Robot.* 2007, 23, 991–1000.
- 20. Gevins, A.; Smith, M.E. Neurophysiological measures of cognitive workload during human-computer interaction. *Theor. Issues Ergon. Sci.* **2003**, *4*, 113–131.
- 21. Van Reekum, C.M.; Johnstone, T.; Banse, R.; Etter, A.; Wehrle, T.; Scherer, K.R. Psychophysiological responses to appraisal dimensions in a computer game. *Cogn. Emotion* **2004**, *18*, 663–668.
- 22. Mandryk, R.L.; Atkins, M.S. A fuzzy physiological approach for continuously modeling emotion during interaction with play technologies. *Int. J. Hum. Comp. Stud.* **2007**, *65*, 329–347.
- Wilson, G.M.; Sasse, M.A. Investigating the Impact of Audio Degradations on Users: Subjective vs. Objective Assessment Methods. In Proceedings of OZCHI 2000: Interfacing Reality in the New Millennium, Sydney, Australia, 4–8 December 2000; pp. 135–142.
- 24. Scheirer, J.; Fernandez, R.; Klein, J.; Picard, R. Frustrating the user on purpose: A step toward building an affective computer. *Interact. Comp.* **2002**, *14*, 93–118.
- 25. Barlett, C.P.; Anderson, C.A.; Swing, E.L. Video game effects—Confirmed, suspected, and speculative. A review of the evidence. *Simul. Gam.* **2009**, *40*, 377–403.
- 26. Hebert, S.; Beland, R.; Dionne-Fournelle, O.; Crete, M.; Lupien, S.J. Physiological stress response to video-game playing: The contribution of built-in music. *Life Sci.* 2005, *76*, 2371–2380.

- Ravaja, N.; Saari, T.; Salminen, M.; Laarni, J.; Holopainen, J.; Järvinen, A. Emotional Response Patterns and Sense of Presence during Video Games: Potential Criterion Variables for Game Design. In Proceedings of the NordiCHI 2004, Tampere, Finland, 26–27 October 2004; pp. 339–347.
- Mandryk, R.L.; Atkins, M.S.; Inkpen, K.M. A Continuous and Objective Evaluation of Emotional Experience with Interactive Play Environments. In Proceedings of the Conference on Human Factors in Computing Systems (CHI 2006), Montreal, QC, Canada, 24–27 April 2006; pp. 1027–1036.
- Kivikangas, M.; Chanel, G.; Cowley, B.; Ekman, I.; Salminen, M.; Järvelä, S.; Ravaja, N. A review of the use of psychophysiological methods in game research. J. Gam. Virt. Worlds 2011, 3, 181–199.
- Kidd, C.D.; Breazeal, C. Human-Robot Interaction Experiments: Lessons Learned. In Proceedings of the AISB'05 Symposium Robot Companions: Hard Problems and Open Challenges in Robot-Human Interaction, Hertfordshire, UK, 14–15 April 2005; pp. 141–142.
- Steinfeld, A.; Fong, T.; Kaber, D.; Lewis, M.; Scholtz, J.; Schultz, A.; Goodrich, M. Common Metrics for Human-Robot Interaction. In Proceedings of the 1st ACM SIGCHI/SIGART Conference on Human-Robot Interaction, Salt Lake City, UT, USA, 2–3 March 2006.
- Bethel, C.L.; Salomon, K.; Burke, J.L.; Murphy, R.R. Psychophysiological Experimental Design for Use in Human-Robot Interaction Studies. In Proceedings of the 2007 International Symposium on Collaborative Technologies and Systems (CTS 2007), Orlando, FL, USA, 25–26 May 2007.
- Borod, J.C.; Bloom, R.L.; Brickman, A.M.; Nakhutina, L.; Curko, E.A. Emotional processing deficits in individuals with unilateral brain damage. Emotional processing deficits in individuals with unilateral brain damage. *Appl. Neuropsychol. Adult* 2002, 9, 23–36.
- 34. Demaree, H.A.; Everhart, D.E.; Youngstrom, E.A.; Harrison, D.W. Brain lateralization of emotional processing: Histrical roots and a future incorporating dominance. *Behav. Cognit. Neurosci. Rev.* **2005**, *4*, 3–20.
- Cacioppo, J.T.; Berntson, G.G.; Larsen, J.T.; Poehlmann, K.M.; Ito, T.A. The psychophysiology of emotion. In *The Handbook of Emotion*, 2nd ed.; Lewis, R., Haviland-Jones, J.M., Eds.; Guilford Press: New York, NY, USA, 2000; pp. 173–191.
- 36. Ruediger, H.; Seibt, R.; Scheuch, K.; Krause, M.; Alam, S.S. Sympathetic and parasympathetic activation in heart rate variability in male hypertensive patients under mental stress. *J. Hum. Hypert.* **2004**, *18*, 307–315.
- 37. Thayer, J.F.; Ahs, F.; Fredrickson, M.; Sollers, J.J.; Wager, T.D. A meta-analysis of heart rate variability and neuroimaging studies: Implications for heart rate variability as a marker of stress and health. *Neurosci. Biobehav. Rev.* **2012**, *36*, 747–756.
- 38. Joyner, J.M.; Charkoudian, N.; Wallin, B.G. A sympathetic view of the sympathetic nervous system and human blood pressure regulation. *Exp. Physiol.* **2008**, *93*, 715–724.
- 39. Acharya, R.U.; Joseph, P.K.; Kannathal, N.; Lim, C.M.; Suri, J.S. Heart rate variability: A review. *Med. Biol. Eng. Comp.* **2006**, *44*, 1031–1051.
- 40. Rainville, P.; Bechara, A.; Naqvi, N.; Damasio, A.R. Basic emotions are associated with distinct patterns of cardiorespiratory activity. *Int. J. Psychophysiol.* **2006**, *61*, 5–18.

- 41. Th érèse, J.M.O.; van de Boxtel, A.; Westerink, D.M. Respiratory sinus arrhythmia responses to induced emotional states: Effects of RSA indices, emotion induction method, age, and sex. *Biol. Psychol.* **2012**, *91*, 128–141.
- 42. Park, B. Psychophysiology as a Tool for HCI Research: Promises and Pitfalls. In Proceedings of the 13th International Conference on Human-Computer Interaction Part I: New Trends, San Diego, CA, USA, 19–24 July 2009; pp. 141–148.
- 43. Wilson, G. Operator functional state assessment for adaptive automation implementation. In *Biomonitoring for Physiological and Cognitive Performance during Military Operations*; Caldwell, J.A., Wesensten, N.J., Eds.; SPIE: Orlando, FL, USA, pp. 100–104.
- 44. Blechert, J.; Lajtman, M.; Michael, T.; Margraf, J.; Wilhelm, F.H. Identifying anxiety states using broad sampling and advanced processing of peripheral physiological information. *Biomed. Sci. Instrum.* **2006**, *42*, 136–141.
- 45. Etzel, J.A.; Johnsen, E.L.; Dickerson, J.A.; Tranel, D.; Adolphs, R. Cardiovascular and respiratory responses during musical mood induction. *Int. J. Psychophysiol.* **2006**, *61*, 57–69.
- 46. Khalfa, S.; Roy, M.; Rainville, P.; Bella, S.D.; Peretz, I. Role of tempo entrainment in psychophysiological differentiation of happy and sad music? *Int. J. Psychophysiol.* **2008**, *68*, 17–26.
- 47. Murakami, H.; Ohira, H. Influence of attention manipulation on emotion and autonomic responses. *Percept. Motor Skills* **2007**, *105*, 299–308.
- 48. Pauls, C.A.; Stemmler, G. Repressive and defensive coping during fear and anger. *Emotion* **2003**, *3*, 284–302.
- 49. Stern, R.M.; Ray, W.J.; Quigley, K.S. *Psychophysiological Recording*, 2nd ed.; Oxford University Press: New York, NY, USA, 2001.
- 50. Benedek, M.; Kaerbach, C. A continuous measure of phasic electrodermal activity. *J. Neurosci. Method* **2010**, *190*, 80–91.
- 51. Dawson, M.E.; Schell, A.M.; Filion, D.L. The electrodermal system. In *Handbook of Psychophysiology*, 2nd ed.; Cacippo, J.T., Tassinary, L.G., Berntson, G.G., Eds.; Cambridge University Press: Cambridge, UK, 2000; pp. 200–223.
- 52. Ritz, T.; Steptoe, A.; Wilde, S.D.; Costa, M. Emotions and stress increase respiratory resistance in asthma. *Psychos. Med.* **2000**, *62*, 401–412.
- 53. Stemmler, G.; Heldmann, M.; Pauls, C.A.; Scherer, T. Constraints for emotion specificity in fear and anger: The context counts. *Psychophysiology* **2001**, *38*, 275–291.
- Haapalainen, E.; Kim, S.J.; Forlizzi, J.F.; Dey, A.K. Psycho-Physiological Measures for Assessing Cognitive Load. In Proceedings of the 12th ACM International Conference on Ubiquitous Computing, Copenhagen, Denmark, 26–29 September 2010; pp. 301–310.
- 55. Shi, Y.; Ruiz, N.; Taib, R.; Choi, E.; Chen, F. *Galvanic Skin Response (GSR) as an Index of Cognitive Load. Ext. Abstracts CHI 2007*; ACM Press: New York, NY, USA, 2007; pp. 2651–2656.
- 56. Raez, M.B.I.; Hussain, M.S.; Mohd-Yasin, F. Techniques of EMG signal analysis: Detection, processing, classification and applications. *Biol. Proced. Online* **2006**, *8*, 11–35.
- 57. Neumann, D.L.; Westbury, H.R. The psychophysiological measurement of empathy. In *Psychology of Empathy*; Scapaletti, D.J., Ed.; Nova Science Publishers Inc.: Hauppauge, NY, USA, 2011; pp. 119–142.

- 58. Dimberg, U.; Thunberg, M.; Grunedal, S. Facial reactions to emotional stimuli: Automatically controlled emotional responses. *Cogn. Emotion* **2002**, *16*, 449–471.
- 59. Wied, M.A.; van de Boxtel, A.; Zaalberg, R.; Goudena, P.P.; Matthys, W.C.H.J. Facial EMG responses to dynamic emotional facial expressions in boys with disruptive behavior disorders. *J. Psychiatr. Res.* **2006**, *40*, 112–121.
- 60. Dimberg, U.; Thunberg, M.; Elmehed, K. Unconscious facial reactions to emotional facial expressions. *Psychol. Sci.* 2000, *11*, 86–89.
- 61. Davidson, R.J.; Pizzagalli, D.; Nitschke, J.B.; Putnam, K.M. Depression: Perspectives from affective neuroscience. *Annu. Rev. Psychol.* **2002**, *53*, 545–574.
- 62. Coan, J.A.; Allen, J.J.B. Frontal EEG asymmetry as a moderator and mediator of emotion. *Biol. Psychol.* **2004**, *67*, 7–49.
- Engels, A.S.; Heller, W.; Mohanty, A.; Herrington, J.D.; Banich, M.T.; Webb, A.G.; Miller, G.A. Specificity of regional brain activity in anxiety types during emotion processing. *Psychophysiology* 2007, 44, 352–363.
- 64. Nitschke, J.B.; Heller, W.; Miller, G.A. Anxiety, Stress, and Cortical Brain Function. In *The Neuropsychology of Emotions*; Borod, J.C., Ed.; Oxford University Press: New York, NY, USA, 2000; pp. 298–319.
- 65. Chaouachi, M.; Jraidi, I.; Frasson, C. Modeling mental workload using eeg features for intelligent systems. User modeling, adaption and personalization. *Lect. Note. Comput. Sci.* **2011**, 6787, 50–60.
- 66. Wilson, G.F. An analysis of mental workload in pilots during flight using multiple psychophysiological measures. *Int. J. Aviat. Psychol.* **2002**, *12*, 3–18.
- 67. Phan, K.L.; Wager, T.D.; Taylor, S.F.; Liberzon, I. Functional brain imaging studies of human emotion. *CNS Spectr.* **2004**, *9*, 258–266.
- 68. Saygin, A.P.; Chaminade, T.; Urgen, B.A.; Ishiguro, H. Cognitive neuroscience and robotics: A mutually beneficial joining of forces. In *Robotics: Systems and Science*; Takayama, L., Ed.; University of Southern California: Los Angeles, CA, USA, 2011.
- 69. Dennett, D. The Intentional Stance; MIT Press: Cambridge, MA, USA, 1987.
- Kim, K.J.; Lipson, H. Towards a Simple Robotic Theory of Mind. In Proceedings of the 9th Workshop on Performance Metrics for Intelligent Systems, PerMIS 2009, Gaithersburg, MD, USA, 21–23 September 2009; pp. 131–138.
- Bonarini, A.; Mainardi, L.; Matteucci, M.; Tognetti, S.; Colombo, R. Stress Recognition in a Robotic Rehabilitation Task. In Proceedings of the Robotic Helpers: User Interaction, Interfaces and Companions in Assistive and Therapy Robotics, Amsterdam, the Netherlands, 12–15 March 2008.
- Kulić, D.; Croft, E.A. Anxiety Detection during Human-Robot Interaction. In Proceedings of the IEEE International Conference on Intelligent Robots and Systems, Edmonton, AB, Canada, 2–6 August 2005; pp. 616–621.
- Kulic, D.; Croft, E. Estimating Intent for Human-Robot Interaction. In Proceedings of the 11th International Conference on Advanced Robotics, Coimbra, Portugal, 30 June–3 July 2003; pp. 810–815.

- Mower, E.; Feil-Seifer, D.J.; Mataric, M.J.; Narayanan, S. Investigating Implicit Cues for User State Estimation in Human-Robot Interaction Using Physiological Measurements. In Proceedings of the 16th IEEE International Symposium on Robot and Human Interactive Communication, RO-MAN 2007, Jeju Island, Korea, 26–29 August 2007; pp. 1125–1130.
- 75. Nasoz, F.; Alvarez, K.; Lisetti, C.L.; Finkelstein, N. Emotion recognition from physiological signals for presence technologies. *Int. J. Cogn. Technol. Work* **2003**, *6*, 1–14.
- 76. Picard, R.W.; Vyzas, E.; Healy, J. Toward machine emotional intelligence: Analysis of affective psychological states. *IEEE Trans. Pattern Anal. Mach. Intell.* **2001**, *23*, 1175–1191.
- 77. Rani, P.; Sims, J.; Brackin, R.; Sarkar, N. Online stress detection using psychophysiological signals for implicit human-robot cooperation. *Robotica* **2002**, *20*, 672–685.
- Rani, P.; Sarkar, N. A new approach to implicit human-robot interaction using affective cues. In *Mobile Robots, towards New Applications*; Lazinica, A., Ed.; InTech: New York, NY, USA, 2006; pp. 233–252.
- Rani, P.; Sarkar, N.; Smith, C.A.; Adams, J.A. Affective Communication for Implicit Human-Machine Interaction. In Proceedings of the IEEE International Conference on Systems, Man, and Cybernetics, Washington, DC, USA, 5–8 October 2003; Volume 5, pp. 4896–4903.
- Zhai, J.; Barreto, A.B.; Chin, C.; Chao, Li. Realization of Stress Detection Using Psychophysiological Signals for Improvement of Human-Computer Interactions. In Proceedings of the IEEE SoutheastCon 2005, Fort Lauderdale, FL, USA, 8–10 April 2005; pp. 415–420.
- Itoh, K.; Miwa, H.; Nukariya, Y.; Zecca, M.; Takanobu, H.; Roccella, S.; Carrozza, M.C.; Dario, P.; Atsuo, T. Development of a Bioinstrumentation System in the Interaction between a Human and a Robot. In Proceedings of the International Conference of Intelligent Robots and Systems, Beijing, China, 9–15 October 2006; pp. 2620–2625.
- Badesa, F.J.; Morales, R.; Garcia-Aracil, N.; Sabater, J.M.; Perez-Vidal, C.; Fernandez, E. Multimodal interfaces to improve therapeutic outcomes in robot-assisted rehabilitation. *IEEE Trans. Syst. Man Cybern. Appl. Rev.* 2012, 42, 1152–1158.
- Guerrero, C.R.; Marinero, J.F.; Turiel, J.P.; Farina, P.R. Using Psychophysiological Feedback to Enhance Physical Human Robot Interaction in a Cooperative Scenario. In Proceedings of the 4th IEEE RAS & EMBS International Conference on Biomedical Robotics and Biomechatronics (BioRob), Roma, Italy, 24–27 June 2012; pp. 1176–1181.
- Koenig, A.; Novak, D.; Omlin, X.; Pulfer, M.; Perreault, E.; Zimmerli, L.; Mihelj, M.; Riener, R. Real-time closed-loop control of cognitive load in neurological patients during robot-assisted gait training. *IEEE Trans. Neural Syst. Rehabil. Eng.* 2011, 19, 453–464.
- 85. Liu, C.; Conn, K.; Sarkar, N.; Stone, W. Online affect detection and robot behavior adaptation for intervention of children with autism. *IEEE Trans. Robot.* **2008**, *24*, 883–896,
- 86. Rani, P.; Sarkar, N.; Smith, C.A.; Kirby, L.D. Anxiety detecting robotic system towards implicit human-robot collaboration. *Robotica* **2004**, *22*, 85–95.
- 87. Ganesh, M. Introduction to Fuzzy Sets and Fuzzy Logic; Prentice Hall of India: New Delhi, India, 2006.
- 88. Dehais, F.; Sisbot, E-A.; Alami, R.; Causse, M. Physiological and subjective evaluation of a human-robot object hand over task. *Appl. Ergon.* **2011**, *42*, 785–791.

- Goljar, N.; Javh, M.; Poje, J.; Ocepek, J.; Novak, D.; Ziherl, J.; Olenšek, A.; Mihelj, M.; Munih, M. Psychophysiological Responses to Robot Training in Different Recovery Phases after Stroke. In Proceedings of the IEEE International Conference on Rehabilitation Robotics (ICORR 2011), Zurich, Switzerland, 29 June–1 July 2011.
- Swangnetr, M.; Kaber, D.B. Emotional state classification in patient-robot interaction using wavelet analysis and statistics-based feature selection. *IEEE Trans. Hum. Mach. Syst.* 2013, 43, 63–75.
- 91. Zhang, T.; Kaber, D.B.; Zhu, B.; Swangnetr, M.; Mosaly, P.; Hodge, L. Service robot feature design effects on user perceptions and emotional responses. *Intell. Serv. Robot.* **2010**, *3*, 73–88.
- 92. Broadbent, E.; Lee, Y.I.; Stafford, R.Q.; Kuo, Y.H.; MacDonald, B.A. Mental schemas of robots as more human-like are associated with higher blood pressure and negative emotions in a human-robot interaction. *Int. J. Soc. Robot* **2011**, *3*, 291–297.
- 93. Watson, D.; Clark, L.A.; Tellegen, A. Development and validation of brief measures of positive and negative affect: The PANAS scales. *J. Pers. Soc. Psychol.* **1988**, *54*, 1063–1070.
- 94. Chaminade, T.; da Fonseca, D.; Rosset, D.; Lutcher, E.; Cheng, G.; Deruelle, C. FMRI Study of Young Adults with Autism Interacting with a Humanoid Robot. In Proceedings of the 21st IEEE International Symposium on Robot and Human Interactive Communication, Paris, France, 9–13 September 2012; pp. 380–385.
- 95. Chaminade, T.; Zecca, M.; Blakemore, S.-J.; Takanishi, A.; Frith, C.D.; Micera, S.; Dario, P.; Rizzolatti, G.; Gallese, V.; Umit à, M.A. Brain response to a humanoid robot in areas implicated in the perception of human emotional gestures. *PLoS One* **2010**, *5*, doi: 10.1371/ journal.pone.0011577.
- 96. Miura, N.; Sugiura, M.; Takahashi, M.; Moridaira, T.; Miyamoto, A.; Kuroki, Y.; Kawashima, R. An Advantage of Bipedal Humanoid Robot on the Empathy Generation: A Neuroimaging Study. In Proceedings of the IEEE/RSJ International Conference on Intelligent Robots and Systems, Nice, France, 22–26 September 2008; pp. 2465–2470.
- 97. Tiberio, L.; Cesta, A.; Cortellessa, G.; Padua, L.; Pellegrino, A.R. Assessing Affective Response of Older Users to a Telepresence Robot Using a Combination of Psychophysiological Measures. In Proceedings of the IEEE RO-MAN: The 21st IEEE International Symposium on Robot and Human Interactive Communication, Paris, France, 9–13 September 2012.
- Cesta, A.; Cortellessa, G.; Orlandini, A.; Tiberio, L. Addressing the Long-Term Evaluation of a Telepresence Robot for the Elderly; Joaquim, F., Ana, L.N., Eds.; SciTePress: Rome, Italy, 2012; pp. 652–663.
- Beer, J.; Takayama, L. Mobile Remote Presence Systems for Older Adults: Acceptance, Benefits, and Concerns. In Proceedings of the 6th International Conference on Human-Robot Interaction, Lausanne, Switzerland, 6–9 March 2011; pp. 19–26.
- 100. Sharkey, A.; Sharkey, N. Children, the elderly, and interactive robots. *IEEE Robot. Autom. Mag.* 2011, *18*, 32–38.
- 101. Giraff. Available online: http://www.giraff.org (accessed on 17 May 2013).
- 102. ExCITE (Enabling SoCial Interaction Through Embodiment). Available online: http://www.oru.se/ excite (accessed on 17 May 2013).

- 103. Spielberger, C.D.; Gorsuch, R.L.; Lushene, R.E. *Manual for the State-Trait Anxiety Inventory*; Consulting Psychologists Press: Palo Alto, CA, USA, 1970.
- 104. Davis, F.D. Perceived usefulness, perceived ease of use, and user acceptance of information technology. *MIS Quart.* **1989**, *13*, 319–340.
- 105. Tapus, A.; Matarić, M.J.; Scassellati, S. The grand challenges in socially assistive robotics. *IEEE Robot. Autom. Mag.* **2007**, *14*, 35–42.
- 106. Kim, K.H.; Bang, S.W.; Kim, S.R. Emotion recognition system using short-term monitoring of physiological signals. *Med. Biol. Eng. Comp.* **2004**, *42*, 419–427.
- 107. Kreibig, S.D. Autonomic nervous system activity in emotion: A review. *Biol. Psychol.* **2010**, *84*, 394–421.
- 108. Minato, T.; Shimada, M.; Itakura, S.; Lee, K.; Ishiguro, H. Evaluating the human likeness of an android by comparing gaze behaviors elicited by the android and a person. *Adv. Robot.* 2006, 20, 1147–1163.
- 109. Mori, M. The uncanny valley. *Energy* **1970**, *74*, 33–35.
- 110. Czaja, S.; Charness, N.; Fisk, A.; Hertzog, C.; Nair, S.; Rogers, W.; Sharit, J. Factors predicting the use of technology: Findings from the center for research and education on aging and technology enhancement. *Psychol. Aging* **2006**, *21*, 333–352.
- 111. Novak, D.; Mihelj, M.; Ziherl, J.; Olensek, A.; Munih, M. Psychophysiological measurements in a biocooperative feedback loop for upper extremity rehabilitation. *IEEE Trans. Neural Syst. Rehabil. Eng.* 2011, 19, 400–410.
- 112. Fairclough, S.H. Fundamentals of physiological computing. Interact. Comp. 2009, 21, 133–145.
- Novak, D.; Mihelj, M.; Munih, M. A survey of methods for data fusion and system adaptation using autonomic nervous system responses in physiological computing. *Interact. Comput.* 2012, 24, 154–172.
- 114. Fisher, R.J. Social desirability bias and the validity of indirect questioning. J. Consum. Res. 1993, 2, 303–315.
- Posner, J.; Russell, J.A.; Peterson, B.S. The circumplex model of affect: An integrative approach to affective neuroscience, cognitive development, and psychopathology. *Dev. Psychopathol.* 2005, *17*, 715–734.

© 2013 by the authors; licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution license (http://creativecommons.org/licenses/by/3.0/).